Author's response to reviews

Title: SPEM Dysfunction and General Schizotypy as Measured by the SSQ: A Controlled Study

Authors:

Dirk van Kampen (d.van.kampen@psy.vu.nl)
Jan Berend Deijen (jb.deijen@psy.vu.nl)

Version: 3 Date: 8 May 2009

Author's response to reviews:

The page numbers 143-156 have been changed in ref 50 into 137-150